Author’s response to reviews

**Title:** PREVALENCE OF GOITER AND ITS ASSOCIATED FACTORS AMONG PRIMARY SCHOOL CHILDREN IN CHOLE DISTRICT, ARSI ZONE, ETHIOPIA: A CROSS SECTIONAL STUDY

**Authors:**

Takele Adilo (admetakele@gmail.com)

Abera Bekele (aberabekele@gmail.com)

**Version:** 6  **Date:** 07 Dec 2018

**Author’s response to reviews:**

The editorial office comments and query regarding the overlap of some of the sentences and concepts from line number 87-102 with a previously done systematic review on the topic were well taken. In addition, as they were occurred unknowingly necessary corrections were done accordingly.